Author’s response to reviews

Title: Social factors in frequent callers: a description of isolation, poverty and quality of life in those calling emergency medical services frequently

Authors:

Gina Agarwal (gina.agarwal@gmail.com)
Janice Lee (janice.lee@medportal.ca)
Brent McLeod (brent.mcleod@hamilton.ca)
Sabnam Mahmuda (mahmudat4@gmail.com)
Michelle Howard (mhoward@mcmaster.ca)
Krista Cockerell (K.Cockrell@westernsydney.edu.au)
Ricardo Angeles (angelesric@gmail.com)

Version: 1 Date: 22 Apr 2019

Author’s response to reviews:

Please find a complete table outlining the point-by-point response to reviewers in a clear format in the supplementary material.